Author’s response to reviews

Title: Trends in maternal and neonatal mortality in South Africa: A systematic review

Authors:

Damian Jeremia Damian (d_jeremy5@yahoo.com)
Bernard Njau (njxber001@myuct.ac.za)
Esther Lisasi (lssest001@myuct.ac.za)
Sia Msuya (siamsuya@hotmail.com)
Andrew Boulle (adrew.boule@uct.ac.za)

Version: 1 Date: 07 Dec 2018

Author’s response to reviews:

1. I need the submission to include page numbers before I will send it out for peer review, please update and resubmit. The line numbers are the same on each page for 32 pages, hence needing page numbers for reference.

The page numbers and line numbers are all updated.